Author's response to reviews

Title: Understanding evidence: a statewide survey to explore evidence-informed public health decision-making in a local government setting

Authors:

Rebecca Armstrong (armr@unimelb.edu.au)
Elizabeth Waters (ewaters@unimelb.edu.au)
Laurence Moore (MooreL1@cf.ac.uk)
Boyd Swinburn (boyd.swinburn@deakin.edu.au)
Maureen Dobbins (dobbinsm@mcmaster.ca)
Laurie Anderson (lmander@uw.edu)
Tahna Pettman (tpettman@unimelb.edu.au)
Cate Burns (cate.m.burns@gmail.com)
Mark Petticrew (Mark.Petticrew@lshtm.ac.uk)

Version: 4 Date: 26 November 2014

Author's response to reviews: see over
26th November 2014

TO: Dr Nick Sevdalis
Implementation Science Editor

Thank you for the opportunity to resubmit this paper with a reduced word count. We have deleted more than 1500 words as requested. We have attached both a track changes and clean copy of the paper so that changes can be viewed. We are confident that the content of the paper in its reduced form reflects the original document’s intent and also responds to the reviewers comments.

We look forward to your favourable response.

Best wishes,

[Signature]

Rebecca Armstrong (on behalf of the author team)